Author's response to reviews

Title: Dermoscopy as a technique for the early identification of foot melanoma

Authors:

Ivan R Bristow (ib@soton.ac.uk)
Jonathan Bowling (jonathan.Bowling@orh.nhs.uk)

Version: 4 Date: 12 March 2009

Author's response to reviews:

Minor, final changes have been made to correct a figure number and correspondence details.